Author’s response to reviews

Title: Isolated Congenital Inter-costal Pulmonary Hernia (ICIPH): A case report

Authors:

Wendlamita Toussaint TAPSOBA (tap_toussaint@yahoo.fr)

Version: 1 Date: 14 Apr 2019

Author’s response to reviews:

Reviewer #1:

Thank you very much

- About the language, I made the requested changes point by point

- For the x-ray, I don't have an other one. Sorry, there was a mistake in the text (x-ray showed no anomaly of the chest and the lung, just a discreet enlargement of the intercostal space). So I didn't check a post-operative picture of x-ray

- about picture a pre-treatment picture of the child (eyes covered) indicating the tumefaction. It don't get it.

During my internship at the Robert Debré hospital in Paris, I physically saw the patient for the first time in the operating room. This is where I asked the surgeon to write an article because it is an exceptional case that I had never seen. The surgeon also did not have a picture taken during the examination of the patient. Sorry

Reviewer #2:

Thank you very much

I made corrections to improve the language. I wanted to translate again but I was short of money. When I had the money, the time to get the translation was not optimal.